Author's response to reviews

Title: Unusually large, segmental, circular vascular malformation of the small intestine in a female toddler with hematochezia

Authors:

Peter I Kalmar (Peter.Kalmar@Medunigraz.at)
Thomas Petnehazy (Thomas.Petnehazy@Medunigraz.at)
Ulrike Wießpeiner (Ulrike.Wiesspeiner@Medunigraz.at)
Meinrad Beer (Meinrad.Beer@Medunigraz.at)
Almuthe C Hauer (Almuthe.Hauer@Medunigraz.at)
Holger Till (Holger.Till@Medunigraz.at)
Michael Riccabona (Michael.Riccabona@Medunigraz.at)

Version: 2
Date: 14 December 2013

Author's response to reviews:

Dear Editor,

Thank you for letting us revise our manuscript. The authors have strengthened all sections with particular focus on the reviewer's comments as well as grammatical and technical details.

Kind regards,

Peter Kalmar